Author’s response to reviews

Title: Predictors of quality of life among inpatients in a forensic hospital. Implications for occupational therapists.

Authors:

Padraic O'Flynn (padraic.oflynn@hse.ie)
Roisin O'Regan (roisin.oregan@hse.ie)
Ken O'Reilly (ken.oreilly1@hse.ie)
Harry Kennedy (kennedh@tcd.ie)

Version: 1 Date: 17 Jul 2017

Author’s response to reviews:

We have now complied with your helpful suggestion that we should closely follow the structure of the STROBE guidelines. We hope the manuscript is now acceptable. We look forward to further responses from reviewers.

We would suggest that amongst those researchers cited in the references section of the manuscript, the following might be appropriate reviewers

Dr Y.H.A. (Yvonne) Bouman y.bouman@transfore.nl

Ellen Vorstenbosch Ellen EV. Vorstenbosch e.vorstenbosch@pssjd.org